Author's response to reviews

Title: Association of Alanine Aminotransferase Levels (ALT) with the Hepatic Insulin Resistance Index (HIRI): a cross-sectional study.

Authors:

Miguel A Gomez-Samano (gsamano83@yahoo.com)
Daniel Cuevas-Ramos (ceptamim@gmail.com)
Roopa Mehta (roopamehta@yahoo.com)
Hasan Brau-Figueroa (hasan.brau@gmail.com)
Clara E Meza-Arana (claraelena88@hotmail.com)
Alfonso Gulias-Herrero (alfonso.guliash@quetzal.innsz.mx)

Version: 5 Date: 27 July 2012

Author's response to reviews:

Dear Timothy Shipley,

We appreciate the revision of our article entitled “Association of Alanine Aminotransferase Levels (ALT) with the Hepatic Insulin Resistance Index (HIRI): a cross-sectional study”.

We have made the correction that reviewer 1 suggested us about the abstract conclusion.

We have revised our ethics statement and provided the name of the ethics committee that approved our study.

Thank you very much.